Author's response to reviews

Title: Do open youth unemployment and youth programs leave the same mental health scars? - Evidence from a Swedish 27-year cohort study

Authors:

Mattias Strandh (mattias.strandh@soc.umu.se)
Karina Nilsson (karina.nilsson@soc.umu.se)
Madelene Nordlund (madelene.nordlund@soc.umu.se)
Anne Hammarström (anne.hammarstrom@umu.se)

Version: 4 Date: 7 November 2015

Author's response to reviews:

Dear Editor,

Thank you for considering our article suitable for publication in the journal. We have now corrected the query that was put to us:

-Could you please check if a year is missing in the first sentence of subsection Methods of the abstract? If so, could you please update both the abstract in the manuscript and the one that is separately uploaded in the submission system?

There was indeed a year missing in the first sentence of subsection Methods of the abstract. This has now been corrected both in the manuscript and in the submission system.

We have also made sure that the manuscript follows the journal style. We here found the provided template very usefull.

Sincerely

Mattias Strandh